Author's response to reviews

Title: Medical students' perceptions of the educational environment at an Iranian Medical Sciences University

Authors:

Teamur Aghamolaei (taghamolaei@hums.ac.ir)
Ismaeil Fazel (ifazel@hums.ac.ir)

Version: 4 Date: 10 November 2010

Author's response to reviews:

Dear Sir/Madam
The BMC Series Editorial Production Team
Article: Medical students' perceptions of the educational environment at an Iranian Medical Sciences University
Reference number: 1068603872355238
Version: 4

Thank you for your email about our manuscript, reference number as 1068603872355238. We have done changes to the format of the manuscript.

Best Wishes

Teamur Aghamolaei, Corresponding author
Department of Public Health, School of Health, Hormozgan University of Medical Sciences, Bandar Abbas, Iran.
Ph: 0098 0761 3338583 Fax: 0098 0761 3338584
E-mail Address: taghamolaei@hums.ac.ir